\begin{thebibliography}{10}

\bibitem{Pand9506:Emerging}
R.~Pandya, ``Emerging mobile and personal communication systems,'' {\em IEEE
  Communications Magazine}, Vol.~33, pp.~44--52, June 1995.

\bibitem{rfc2205}
R.~Braden, Ed., L.~Zhang, S.~Berson, S.~Herzog, and S.~Jamin, ``Resource
  {ReSerVation} protocol {(RSVP)} -- version 1 functional specification,''
  Request for Comments 2205, Internet Engineering Task Force, Sept. 1997.

\bibitem{rfc1889}
H.~Schulzrinne, S.~Casner, R.~Frederick, and V.~Jacobson, ``{RTP:} a transport
  protocol for real-time applications,'' Request for Comments 1889, Internet
  Engineering Task Force, Jan. 1996.

\bibitem{rfc2326}
H.~Schulzrinne, A.~Rao, and R.~Lanphier, ``Real time streaming protocol
  {(RTSP)},'' Request for Comments 2326, Internet Engineering Task Force, Apr.
  1998.

\bibitem{rfc2974}
M.~Handley, C.~Perkins, and E.~Whelan, ``Session announcement protocol,''
  Request for Comments 2974, Internet Engineering Task Force, Oct. 2000.

\bibitem{rfc2327}
M.~Handley and V.~Jacobson, ``{SDP:} session description protocol,'' Request
  for Comments 2327, Internet Engineering Task Force, Apr. 1998.

\bibitem{rfc2119}
S.~Bradner, ``Key words for use in {RFCs} to indicate requirement levels,''
  Request for Comments 2119, Internet Engineering Task Force, Mar. 1997.

\bibitem{rfc2616}
R.~Fielding, J.~Gettys, J.~Mogul, H.~Frystyk, L.~Masinter, P.~Leach, and
  T.~Berners-Lee, ``Hypertext transfer protocol -- {HTTP/1.1},'' Request for
  Comments 2616, Internet Engineering Task Force, June 1999.

\bibitem{rfc2396}
T.~Berners-Lee, R.~Fielding, and L.~Masinter, ``Uniform resource identifiers
  {(URI):} generic syntax,'' Request for Comments 2396, Internet Engineering
  Task Force, Aug. 1998.

\bibitem{rfc1738}
T.~Berners-Lee, L.~Masinter, and M.~McCahill, ``Uniform resource locators
  {(URL)},'' Request for Comments 1738, Internet Engineering Task Force, Dec.
  1994.

\bibitem{draft-ietf-sip-dhcp}
G.~Nair and H.~Schulzrinne, ``{DHCP} option for {SIP} servers,'' {Internet
  Draft}, Internet Engineering Task Force, Apr. 2000.
\newblock Work in progress.

\bibitem{rfc2782}
A.~Gulbrandsen, P.~Vixie, and L.~Esibov, ``A {DNS} {RR} for specifying the
  location of services {(DNS} {SRV)},'' Request for Comments 2782, Internet
  Engineering Task Force, Feb. 2000.

\bibitem{rfc2219}
M.~Hamilton and R.~Wright, ``Use of {DNS} aliases for network services,''
  Request for Comments 2219, Internet Engineering Task Force, Oct. 1997.

\bibitem{rfc1035}
P.~V. Mockapetris, ``Domain names - implementation and specification,'' Request
  for Comments 1035, Internet Engineering Task Force, Nov. 1987.

\bibitem{rfc1288}
D.~Zimmerman, ``The finger user information protocol,'' Request for Comments
  1288, Internet Engineering Task Force, Dec. 1991.

\bibitem{rfc2167}
S.~Williamson, M.~Kosters, D.~Blacka, J.~Singh, and K.~Zeilstra, ``Referral
  whois (rwhois) protocol {V1.5},'' Request for Comments 2167, Internet
  Engineering Task Force, June 1997.

\bibitem{rfc1777}
W.~Yeong, T.~Howes, and S.~Kille, ``Lightweight directory access protocol,''
  Request for Comments 1777, Internet Engineering Task Force, Mar. 1995.

\bibitem{Scho9608:Multicast}
E.~M. Schooler, ``A multicast user directory service for synchronous
  rendezvous,'' Master's Thesis CS-TR-96-18, Department of Computer Science,
  California Institute of Technology, Pasadena, California, Aug. 1996.

\bibitem{rfc2806}
A.~Vaha-Sipila, ``{URLs} for telephone calls,'' Request for Comments 2806,
  Internet Engineering Task Force, Apr. 2000.

\bibitem{rfc768}
J.~Postel, ``User datagram protocol,'' Request for Comments 768, Internet
  Engineering Task Force, Aug. 1980.

\bibitem{rfc761}
J.~Postel, ``{DoD} standard transmission control protocol,'' Request for
  Comments 761, Internet Engineering Task Force, Jan. 1980.

\bibitem{rfc2246}
T.~Dierks and C.~Allen, ``The {TLS} protocol version 1.0,'' Request for
  Comments 2246, Internet Engineering Task Force, Jan. 1999.

\bibitem{rfc2279}
F.~Yergeau, ``{UTF-8,} a transformation format of {ISO} 10646,'' Request for
  Comments 2279, Internet Engineering Task Force, Jan. 1998.

\bibitem{Stev94:TCP1}
W.~R. Stevens, {\em {TCP/IP} illustrated: the protocols}, Vol.~1.
\newblock Reading, Massachusetts: Addison-Wesley, 1994.

\bibitem{rfc1191}
J.~C. Mogul and S.~E. Deering, ``Path {MTU} discovery,'' Request for Comments
  1191, Internet Engineering Task Force, Nov. 1990.

\bibitem{rfc822}
D.~Crocker, ``Standard for the format of {ARPA} internet text messages,''
  Request for Comments 822, Internet Engineering Task Force, Aug. 1982.

\bibitem{rfc2365}
D.~Meyer, ``Administratively scoped {IP} multicast,'' Request for Comments
  2365, Internet Engineering Task Force, July 1998.

\bibitem{rfc1890}
H.~Schulzrinne, ``{RTP} profile for audio and video conferences with minimal
  control,'' Request for Comments 1890, Internet Engineering Task Force, Jan.
  1996.

\bibitem{rfc1750}
D.~Eastlake, 3rd, S.~Crocker, and J.~Schiller, ``Randomness recommendations for
  security,'' Request for Comments 1750, Internet Engineering Task Force, Dec.
  1994.

\bibitem{rfc2426}
F.~Dawson and T.~Howes, ``vcard {MIME} directory profile,'' Request for
  Comments 2426, Internet Engineering Task Force, Sept. 1998.

\bibitem{rfc2849}
G.~Good, ``The {LDAP} data interchange format {(LDIF)} - technical
  specification,'' Request for Comments 2849, Internet Engineering Task Force,
  June 2000.

\bibitem{rfc2368}
P.~Hoffman, L.~Masinter, and J.~Zawinski, ``The mailto {URL} scheme,'' Request
  for Comments 2368, Internet Engineering Task Force, July 1998.

\bibitem{rfc1806}
R.~Troost and S.~Dorner, ``Communicating presentation information in internet
  messages: The content-disposition header,'' Request for Comments 1806,
  Internet Engineering Task Force, June 1995.

\bibitem{rfc1123}
R.~T. Braden, ``Requirements for internet hosts - application and support,''
  Request for Comments 1123, Internet Engineering Task Force, Oct. 1989.

\bibitem{rfc2076}
J.~Palme, ``Common internet message headers,'' Request for Comments 2076,
  Internet Engineering Task Force, Feb. 1997.

\bibitem{rfc1321}
R.~Rivest, ``The {MD5} message-digest algorithm,'' Request for Comments 1321,
  Internet Engineering Task Force, Apr. 1992.

\bibitem{rfc2277}
H.~Alvestrand, ``{IETF} policy on character sets and languages,'' Request for
  Comments 2277, Internet Engineering Task Force, Jan. 1998.

\bibitem{rfc2046}
N.~Freed and N.~Borenstein, ``Multipurpose internet mail extensions {(MIME)}
  part two: Media types,'' Request for Comments 2046, Internet Engineering Task
  Force, Nov. 1996.

\bibitem{rfc1991}
D.~Atkins, W.~Stallings, and P.~Zimmermann, ``{PGP} message exchange formats,''
  Request for Comments 1991, Internet Engineering Task Force, Aug. 1996.

\bibitem{rfc1825}
R.~Atkinson, ``Security architecture for the internet protocol,'' Request for
  Comments 1825, Internet Engineering Task Force, Aug. 1995.

\bibitem{rfc2617}
J.~Franks, P.~Hallam-Baker, J.~Hostetler, S.~Lawrence, P.~Leach, A.~Luotonen,
  and L.~Stewart, ``{HTTP} authentication: Basic and digest access
  authentication,'' Request for Comments 2617, Internet Engineering Task Force,
  June 1999.

\bibitem{rfc2440}
J.~Callas, L.~Donnerhacke, H.~Finney, and R.~Thayer, ``{OpenPGP} message
  format,'' Request for Comments 2440, Internet Engineering Task Force, Nov.
  1998.

\bibitem{rfc2045}
N.~Freed and N.~Borenstein, ``Multipurpose internet mail extensions {(MIME)}
  part one: Format of internet message bodies,'' Request for Comments 2045,
  Internet Engineering Task Force, Nov. 1996.

\bibitem{rfc2234}
D.~Crocker, Ed., and P.~Overell, ``Augmented {BNF} for syntax specifications:
  {ABNF},'' Request for Comments 2234, Internet Engineering Task Force, Nov.
  1997.

\bibitem{rfc2732}
R.~Hinden, B.~Carpenter, and L.~Masinter, ``Format for literal {IPv6} addresses
  in {URL's},'' Request for Comments 2732, Internet Engineering Task Force,
  Dec. 1999.

\bibitem{Scho9306:Case}
E.~M. Schooler, ``Case study: multimedia conference control in a
  packet-switched teleconferencing system,'' {\em Journal of Internetworking:
  Research and Experience}, Vol.~4, pp.~99--120, June 1993.
\newblock ISI reprint series ISI/RS-93-359.

\bibitem{Schu9603:Personal}
H.~Schulzrinne, ``Personal mobility for multimedia services in the
  {Internet},'' in {\em European Workshop on Interactive Distributed Multimedia
  Systems and Services (IDMS)}, (Berlin, Germany), Mar. 1996.

\end{thebibliography}

\section{SIP Message Overview}
\label{sec:message}

SIP is a text-based protocol and uses the ISO 10646 character set in
UTF-8 encoding (RFC 2279 \cite{rfc2279}).  Senders {\MUST} terminate
lines with a CRLF, but receivers {\MUST} also interpret CR and LF by
themselves as line terminators.  Only the combinations CR CR, LF LF and
CRLF CRLF terminate the message header.  Implementations {\MUST} only
send CRLF CRLF.  \motivation{CR and LF instead of CRLF is for
backwards-compatibility; their use is deprecated.}

Except for the above difference in character sets and line termination,
much of the message syntax is and header fields are identical to
HTTP/1.1; rather than repeating the syntax and semantics here we use
[HX.Y] to refer to Section X.Y of the current HTTP/1.1 specification
(RFC 2616 \cite{rfc2616}).  In addition, we describe SIP in both prose
and an augmented Backus-Naur form (ABNF).  See section \ref{sec:abnf}
for an overview of ABNF.

Note, however, that SIP is not an extension of HTTP.

Unlike HTTP, SIP {\MAY} use UDP or other unreliable datagram protocols. 
Each such datagram carries one request or response.  Datagrams,
including all headers, {\SHOULDNOT} be larger than the path maximum
transmission unit (MTU) if the MTU is known, or 1500 bytes if the MTU is
unknown.  However, implementations {\MUST} be able to handle messages up
to the maximum datagram packet size.  For UDP, this size is 65,535
bytes, including headers.

\motivation{The MTU of 1500 bytes accommodates encapsulation within the
``typical'' ethernet MTU without IP fragmentation.  Recent studies
\cite[p.~154]{Stev94:TCP1} indicate that an MTU of 1500 bytes is a
reasonable assumption.  The next lower common MTU values are 1006 bytes
for SLIP and 296 for low-delay PPP (RFC 1191 \cite{rfc1191}).  Thus,
another reasonable value would be a message size of 950 bytes, to
accommodate packet headers within the SLIP MTU without fragmentation.}

A SIP message is either a request from a client to a server, or a
response from a server to a client.

\begin{syntax}
SIP-message & = & Request $|$ Response
\end{syntax}

Both \header{Request} (section \ref{sec:request}) and \header{Response}
(section \ref{sec:response}) messages use the \header{generic-message}
format of RFC 822 \cite{rfc822} for transferring entities (the body of
the message).  Both types of messages consist of a \header{start-line},
one or more header fields (also known as ``headers''), an empty line
(i.e., a line with nothing preceding the carriage-return line-feed
(\header{CRLF})) indicating the end of the header fields, and an
optional \header{message-body}.  To avoid confusion with similar-named
headers in HTTP, we refer to the headers describing the message body as
entity headers.  These components are described in detail in the
upcoming sections.

\begin{syntax}
generic-message & = & start-line\\
                &   & *message-header\\
                &   & CRLF\\
                &   & [ message-body ]\\
&&\\
start-line      & = & Request-Line $|$ & {\rm ;Section~\ref{sec:Request-Line}}\\
                &   & Status-Line      & {\rm ;Section~\ref{sec:Status-Line}}\\ 
\end{syntax}

\begin{syntax}
message-header
& = & ( general-header     & \\
&   & $|$ request-header   & \\
&   & $|$ response-header  & \\
&   & $|$ entity-header )  & \\
\end{syntax}

\begin{changebar}
\begin{table}
\begin{syntax}
\header{general-header}
&  =  & \header{Accept}                & ; Section~\ref{sec:Accept}\\
& $|$ & \header{Accept-Encoding}       & ; Section~\ref{sec:Accept-Encoding}\\
& $|$ & \header{Accept-Language}       & ; Section~\ref{sec:Accept-Language}\\
& $|$ & \header{Call-ID}               & ; Section~\ref{sec:Call-ID}\\
& $|$ & \header{Call-Info}             & ; Section~\ref{sec:Call-Info}\\
& $|$ & \header{Contact}               & ; Section~\ref{sec:Contact}\\
& $|$ & \header{CSeq}                  & ; Section~\ref{sec:CSeq}\\
& $|$ & \header{Date}                  & ; Section~\ref{sec:Date}\\
& $|$ & \header{Encryption}            & ; Section~\ref{sec:Encryption}\\
& $|$ & \header{From}                  & ; Section~\ref{sec:From}\\
& $|$ & \header{MIME-Version}          & ; Section~\ref{sec:MIME-Version}\\
& $|$ & \header{Organization}          & ; Section~\ref{sec:Organization}\\
& $|$ & \header{Record-Route}          & ; Section~\ref{sec:Record-Route}\\
& $|$ & \header{Require}               & ; Section~\ref{sec:Require}\\
& $|$ & \header{Supported}             & ; Section~\ref{sec:Supported}\\
& $|$ & \header{Timestamp}             & ; Section~\ref{sec:Timestamp}\\
& $|$ & \header{To}                    & ; Section~\ref{sec:To}\\
& $|$ & \header{User-Agent}            & ; Section~\ref{sec:User-Agent}\\
& $|$ & \header{Via}                   & ; Section~\ref{sec:Via}\\
\header{entity-header}
&  =  & \header{Allow}                 & ; Section~\ref{sec:Allow}\\
& $|$ & \header{Content-Disposition}   & ; Section~\ref{sec:Content-Disposition}\\
& $|$ & \header{Content-Encoding}      & ; Section~\ref{sec:Content-Encoding}\\
& $|$ & \header{Content-Language}      & ; Section~\ref{sec:Content-Language}\\
& $|$ & \header{Content-Length}        & ; Section~\ref{sec:Content-Length}\\
& $|$ & \header{Content-Type}          & ; Section~\ref{sec:Content-Type}\\
& $|$ & \header{Expires}               & ; Section~\ref{sec:Expires}\\
\header{request-header}
& $=$ & \header{Alert-Info}            & ; Section~\ref{sec:Alert-Info}\\
& $|$ & \header{Authorization}         & ; Section~\ref{sec:Authorization}\\
& $|$ & \header{In-Reply-To}           & ; Section~\ref{sec:In-Reply-To}\\
& $|$ & \header{Max-Forwards}          & ; Section~\ref{sec:Max-Forwards}\\
& $|$ & \header{Priority}              & ; Section~\ref{sec:Priority}\\
& $|$ & \header{Proxy-Authorization}   & ; Section~\ref{sec:Proxy-Authorization}\\
& $|$ & \header{Proxy-Require}         & ; Section~\ref{sec:Proxy-Require}\\
& $|$ & \header{Route}                 & ; Section~\ref{sec:Route}\\
& $|$ & \header{Response-Key}          & ; Section~\ref{sec:Response-Key}\\
& $|$ & \header{Subject}               & ; Section~\ref{sec:Subject}\\
\header{response-header}
& $=$ & \header{Authentication-Info}   & ; Section~\ref{sec:Authentication-Info}\\
& $|$ & \header{Error-Info}            & ; Section~\ref{sec:Error-Info}\\
& $|$ & \header{Proxy-Authenticate}    & ; Section~\ref{sec:Proxy-Authenticate}\\
& $|$ & \header{Retry-After}           & ; Section~\ref{sec:Retry-After}\\
& $|$ & \header{Server}                & ; Section~\ref{sec:Server}\\
& $|$ & \header{Unsupported}           & ; Section~\ref{sec:Unsupported}\\
& $|$ & \header{Warning}               & ; Section~\ref{sec:Warning}\\
& $|$ & \header{WWW-Authenticate}      & ; Section~\ref{sec:WWW-Authenticate}\\
\end{syntax}
\caption{SIP headers}
\label{tab:sip-headers}
\end{table}
\end{changebar}

In the interest of robustness, any leading empty line(s) {\MUST} be
ignored.  In other words, if the \header{Request} or \header{Response}
message begins with one or more \header{CRLF}, \header{CR}, or
\header{LF}s, these characters {\MUST} be ignored.

\newpage
